Author's response to reviews

Title: Dietary habits in three Central and Eastern European countries: the HAPIEE study

Authors:

Sinead Boylan (s.boylan@public-health.ucl.ac.uk)
Ailsa Welch (A.Welch@uea.ac.uk)
Hynek Pikhart (h.pikhart@ucl.ac.uk)
Sofia Malyutina (smalyutina@hotmail.com)
Andrzej Pajak (mmpajak@cyf-kr.edu.pl)
Ruzena Kubinova (kubinova@zu.cz)
Oksana Bragina (bragina_ok@mail.ru)
Galina Simonova (simonova@iimed.ru)
Urszula Stepaniak (mxbiela@cyf-kr.edu.pl)
Alessandra Gilis-Januszewska (myjanusz@cyf-kr.edu.pl)
Lubomira Milla (luba.milla@seznam.cz)
Anne Peasey (a.peasey@ucl.ac.uk)
Michael Marmot (m.marmot@ucl.ac.uk)
Martin Bobak (m.bobak@ucl.ac.uk)

Version: 3 Date: 6 November 2009

Author's response to reviews:

06/11/09

Re: Response to referee/editor

Dear Editor,

We are delighted that you wish to accept our manuscript MS: 5716663062899822 “Dietary habits in three Central and Eastern European countries: the HAPIEE study” for publication. We would like to thank the referees for their report and have revised the paper accordingly.

We hope that we have addressed all issues raised by the reviewers and by the editor.

Yours sincerely,
Sinead Boylan.

RESPONSE TO REVISIONS

Abstract, Methods.
"aged 45-49 years old" should be "aged 45-49 years" or simply "aged 45-49":
Corrected.
P. 6, para 2.
"aged 45-49 years old". The same correction to be made: Corrected.
P. 11, para 1.
"from 0-7" should be "from 0 to 7": Corrected.
"fruit and vegetables" should be "fruits and vegetables": Corrected.